Author’s response to reviews

Title: Towards personalised integrated dementia care: a qualitative study into the implementation of different models of casemagement.

Authors:

Lisa D van Mierlo (l.vanmierlo@vumc.nl)
Franka JM Meiland (fj.meiland@vumc.nl)
Hein PJ van Hout (hpi.vanhout@vumc.nl)
Rose-Marie Dröes (rm.droes@vumc.nl)

Version: 6
Date: 14 January 2014

Author's response to reviews: see over
To the editor of
BMC Geriatrics

Concerns: Paper for publication in BMC Geriatrics,

Dear Professor Irene Pala,

Please find attached the revised paper “Towards personalised integrated dementia care: a qualitative study on facilitators and barriers to the implementation of two case management models”.

We corrected the original paper based on the comments of the two reviewers.

We would be very pleased if you would reconsider this paper for publication in BMC Geriatrics

Awaiting your answer,
yours sincerely,

Lisa van Mierlo, MSc